Author's response to reviews

Title: Epidermal Growth Factor Receptor (EGFR) mutation analysis, gene expression profiling and EGFR protein expression in androgen-dependent prostate cancer

Authors:

Caterina Peraldo-Neia (caterina.peraldoneia@ircc.it)
Giorgia Migliardi (giorgia.migliardi@ircc.it)
Maurizia Mello-Grand (Maurizia.mellogrand@fondoedotempia.it)
Filippo Montemurro (filippo.montemurro@ircc.it)
Raffaella Segir (raffaella.segir@gmail.com)
Ymera Pignochino (ymera.pignochino@ircc.it)
Giuliana Cavalloni (giuliana.cavalloni@ircc.it)
Bruno Torchio (btorchio@mauriziano.it)
Luciano Mosso (lmosso@mauriziano.it)
Giovanna Chiorino (giovanna.chiorino@fondoedotempia.it)
Massimo Aglietta (massimo.aglietta@ircc.it)

Version: 3 Date: 19 January 2011

Author's response to reviews:

MS: 1166528067447649
Research article
Epidermal Growth Factor Receptor (EGFR) mutation analysis, gene expression profiling and EGFR protein expression in androgen-dependent prostate cancer

Dear Editor,

We resubmit the manuscript to your attention after the English revision made by a native English speaker as you and reviewers suggested.
All changes in the text are highlighted in gray.

We hope that after the English revision our paper will be acceptable for publication on BMC Cancer.